Author’s response to reviews

Title: Epidemiological and Clinical Features of Human Rabies Cases in Bali 2008-2010

Authors:

Ni M Susilawathi (susilawathi.drmade@yahoo.com)
Agus E Darwinata (darwinsync8@gmail.com)
Ida BNP Dwija (bio_dc9814@yahoo.com)
Nyoman S Budayanti (nynsri@yahoo.com)
Gusti AK Wirasandhi (ken_wirasandi@yahoo.co.id)
Ketut Subrata (k_brata@yahoo.com)
Ni K Susilarini (niketutsusi@yahoo.com)
Raka AA Sudewi (raka_sudewi2006@yahoo.com)
Steve Wignall (suwitno@mac.com)
Gusti NK Mahardika (gnmahardika@indosat.net.id)

Version: 2 Date: 6 September 2011

Author's response to reviews:

Dear Madam/Sir

Herewith we have revised our manuscript according to your suggestion. We have added an additional sub-heading in the material and method as the following.

Ethical Approval

The families of all patients agreed that samples could be collected and that the patient data could be included in research and have signed an informed consent form. Ethical clearance for this study has been approved by Research Ethics Committee of the Faculty of Medicine, Udayana University, Denpasar, Bali, Indonesia, reference number 723/Skrt/XI/2010 dated November 12th, 2010.

Therefore we would like to ask for further review process.

Sincerely Yours

GNK Mahardika